Author's response to reviews

**Title:** Polymorphisms near EXOC4 and LRGUK on chromosome 7q32 are associated with Type 2 Diabetes and fasting glucose; The NHLBI Family Heart Study

**Authors:**

Jason M Laramie (laramiej@gmail.com)
Jemma B Wilk (jwilk@bu.edu)
Sally L Williamson (sallyw@bu.edu)
Michael W Nagle (mwn@bu.edu)
Jeanne C Latourelle (jlatoure@bu.edu)
Jennifer E Tobin (jetobin@bu.edu)
Michael A Province (mike@dsgmail.wustl.edu)
Ingrid B Borecki (ingrid@dsgmail.wustl.edu)
Richard H Myers (rmyers@bu.edu)

**Version:** 5 **Date:** 9 May 2008

**Author's response to reviews:**

Added Country in the affiliation details.

Keywords were removed

Author contributions was changed to Authors' contributions

Manuscript sections we changed to Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any);

The reference list now contains all authors names.

Figure 1 was cropped to remove whitespace